Author's response to reviews

Title: Multiple cavernous hemangiomas of the skull with dural tail sign: A case report and literature review

Authors:

Xu Peng (xupenga911@163.com)
Lan Shengyong (gxlansyu@163.com)
Liang Youming (liang_youming@sohu.com)
Xiao Quan (xiaobsh@yahoo.com.cn)

Version: 2 Date: 23 April 2013

Author's response to reviews:

1. We confirm that the patient has given their consent for publication of the case report. Several sentences have been added following the Discussion section.
2. We have formatted the abstract according to the guidelines for authors, and replaced the original content.
3. We have added the Authors' Contributions section after the Competing Interests section.
4. We deleted the list of authors from the title page, and added the name to the 'first name' box and the bracketed abbreviation to the 'last name’ box in the submission system.